



University of Groningen

# J-Inner-Outer Factorization, J-Spectral Factorization, and Robust Control for Nonlinear Systems

Ball, Joseph A.; Schaft, Arjan J. van der

Published in: IEEE Transactions on Automatic Control

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 1996

Link to publication in University of Groningen/UMCG research database

*Citation for published version (APA):* Ball, J. A., & Schaft, A. J. V. D. (1996). J-Inner-Outer Factorization, J-Spectral Factorization, and Robust Control for Nonlinear Systems. *IEEE Transactions on Automatic Control*.

## Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

### Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.

# J-Inner–Outer Factorization, J-Spectral Factorization, and Robust Control for Nonlinear Systems

Joseph A. Ball, Member, IEEE, and Arjan J. van der Schaft, Member, IEEE

Abstract—The problem of expressing a given nonlinear state-space system as the cascade connection of a lossless system and a stable, minimum-phase system (inner-outer factorization) is solved for the case of a stable system having state-space equations affine in the inputs. The solution is given in terms of the stabilizing solution of a certain Hamilton–Jacobi equation. The stable, minimum-phase factor is obtained as the solution of an associated nonlinear spectral factorization problem. As an application, one can arrive at the solution of the nonlinear  $H_{\infty}$ -control problem for the disturbance feedforward case.

## I. INTRODUCTION

T is well known that inner-outer [i.e., lossless (stable, minimum-phase)] and spectral factorization plays a prominent role in  $H_{\infty}$ -control theory (see [7], [14], [15], [18], [19], and [21] for the linear case and [9] for the nonlinear case) as well as in chemical process control (see [13] and [30]). Actually the  $H_{\infty}$ -control application calls more generally for J-inner-outer factorization, where J is an indefinite signature matrix. While computation of both inner-outer and spectral factorization in terms of state-space realizations is well developed in the linear case, the issues are not so well understood in the nonlinear setting. In particular, the nonlinear inner-outer factorization problem was formulated and studied in [2]-[4] for discrete time systems. Specifically, the inner-outer factorization problem for a stable, invertible, discrete-time system was solved in [4] by constructing an invertible, lossless system having a given zero dynamics in terms of the solution of a certain Hamilton-Jacobi equation. Here we recover a version of the same result, but for stable, invertible, continuous-time systems having state-space equations which are affine in the inputs via a different approach. More precisely, we formulate and solve (again in terms of the solution of a Hamilton-Jacobi equation) a nonlinear spectral factorization problem; this leads directly to the computation of the outer factor in an inner-outer factorization of a given stable plant. Thus we obtain the inner-outer factorization by first solving for the outer factor, rather than in the reverse order as is done in [4]. The solution of the nonlinear spectral factorization problem is modeled on the approach in [10] for the linear case and also has contact with

Publisher Item Identifier S 0018-9286(96)02103-4.

the theory of adjoint systems and input-output Hamiltonian systems developed in [12].

Section II introduces the problem in precise form (both in an input-output and state-space setting) and presents the solution. Section III presents as an application an alternative derivation of the Hamilton-Jacobi equation associated with a system having  $L_2$ -gain equal to, at most, a prescribed level  $\gamma$ . Section IV presents the application to nonlinear  $H_{\infty}$ -control theory, and Section V discusses the difficulties of our approach for the unstable case. In a companion paper [27], we plan to discuss the special features of the definite case (where J is the identity matrix), the case of noninvertible plants, and the applications to chemical process control. Finally we mention that an announcement of these results appears in [25] and [26].

#### II. J-INNER-OUTER FACTORIZATION

#### A. Preliminaries

We consider a smooth  $(C^{\infty})$  nonlinear state-space system which is affine in the inputs u

$$\Sigma: \begin{cases} \dot{x} = a(x) + b(x)u, & u \in \mathbf{R}^m \\ y = c(x) + d(x)u, & y \in \mathbf{R}^p \end{cases}$$
(1)

with outputs y, where  $x = (x_1, \dots, x_n)$  are local coordinates for a state-space manifold M. Throughout, we assume the existence of an equilibrium  $x_0$ , i.e.,  $a(x_0) = 0$ . Without loss of generality we also assume that  $c(x_0) = 0$ . Furthermore we make the following standing assumption:

$$x_0 = 0$$
 is a globally asymptotically stable equilibrium  
point of  $\dot{x} = a(x)$ . (A1)

(We remark on the unstable case in Section V.) Because of Assumption (A1), we may assume that the state-space manifold M is equal to  $\mathbb{R}^n$ , and after a suitable coordinate shift, we may take the equilibrium point  $x_0$  to be equal to  $0 \in \mathbb{R}^n$ . In the sequel we will always abbreviate "globally asymptotically stable" to "asymptotically stable." Assume also that we are given an  $m \times m$  signature matrix j and a  $p \times p$  signature matrix J. In general we say that a square matrix  $\mathcal{J}$  is a signature matrix if  $\mathcal{J} = \mathcal{J}^* = \mathcal{J}^{-1}$ , for example,  $\mathcal{J} = \begin{bmatrix} I_{n+} & 0 \\ 0 & -I_{n-} \end{bmatrix}$ . The problem of (j, J)-inner-outer factorization of  $\Sigma$  consists of constructing (if possible) a (j, J)-lossless system  $\Theta$  (the (j, J)-inner factor) and an asymptotically stable minimum

0018-9286/96\$05.00 © 1996 IEEE

Manuscript received November 4, 1994; revised May 10, 1995. Recommended by Associate Editor, A. M. Bloch.

J. A. Ball is with the Department of Mathematics, Virginia Polytechnic Institute and State University, Blacksburg, VA 24061-0123 USA.

A. J. van der Schaft is with the Faculty of Applied Mathematics, University of Twente, Enschede, The Netherlands.

phase system R (the outer factor), both of the same form as  $\Sigma,$  such that

$$\Sigma = \Theta \circ R. \tag{2}$$

By this we mean that for every initial condition of  $\Sigma$ , there exist initial conditions of  $\Theta$  and R such that the input-output map of  $\Sigma$  equals the input-output map of the series interconnection of R followed by  $\Theta$  (for the respective initial conditions). Let us recall (see [29]) that a nonlinear system  $\Sigma$  is called lossless with respect to the supply rate  $\frac{1}{2}u^Tju - \frac{1}{2}y^TJy$ , if there exists a function  $V(x) \geq 0$  (the storage function) such that

$$V(x(t_1)) - V(x(t_0)) = \frac{1}{2} \int_{t_0}^{t_1} (u(t)^T j u(t) - y(t)^T J y(t)) dt$$
$$V(0) = 0$$
(3)

for all  $t_0 \leq t_1$  and  $u(\cdot)$ , with  $x(t_1)$  denoting the state at time  $t_1$  resulting from initial state  $x(t_0)$  at time  $t_0$  and input  $u(\cdot)$  on the time interval  $[t_0, t_1]$ . If V is differentiable, then (3) can be equivalently expressed as

$$V_x(x)[a(x) + b(x)u] = \frac{1}{2}u^T ju - \frac{1}{2}[c(x) + d(x)u]^T J[c(x) + d(x)u]$$
  
$$V(0) = 0$$

for all x, u, or equivalently, as the system of equations

$$V_{x}(x)a(x) + \frac{1}{2}c^{T}(x)Jc(x) = 0, \qquad V(0) = 0$$
  

$$V_{x}(x)b(x) + c^{T}(x)Jd(x) = 0$$
  

$$d^{T}(x)Jd(x) = j \qquad (4)$$

for all x. Here  $V_x(x)$  denotes the row vector of partial derivatives  $[\partial V/\partial x_1(x)\cdots \partial V/\partial x_n(x)]$ .

In this paper we consider minimum phase systems only for the case of causally invertible systems, i.e., a system of the form of (1) with d(x) an invertible matrix for all x; the factorization problem in the singular case will be discussed in a separate paper [27]. In general, if R is a smooth state-space system

$$R: \begin{cases} \dot{x} = \overline{a}(x) + \overline{b}(x)u, & u \in \mathbf{R}^m\\ \overline{y} = \overline{c}(x) + \overline{d}(x)u, & \overline{y} \in \mathbf{R}^m \end{cases}$$
(5)

with  $\overline{d}(x)$  an invertible  $m \times m$  matrix for all  $x \in \mathbb{R}^n$ , then the inverse of R (the system with the same set of trajectories (x(t), u(t), y(t)) but with  $\overline{y}(t)$  appearing as the input variable and u(t) as the output variable) is given explicitly as

$$R^{-1}: \begin{cases} \dot{x} = [\overline{a}(x) - \overline{b}(x)\overline{d}(x)^{-1}\overline{c}(x)] + \overline{b}(x)\overline{d}(x)^{-1}\overline{y} \\ u = -\overline{d}(x)^{-1}\overline{c}(x) + \overline{d}(x)^{-1}\overline{y}. \end{cases}$$
(6)

We say that a system R, as in (5), is minimum phase if the dynamics of the inverse system is Lyapunov stable, i.e., if zero is a stable equilibrium point in the sense of Lyapunov for the system of differential equations

$$\dot{x} = \overline{a}(x) - \overline{b}(x)\overline{d}(x)^{-1}\overline{c}(x).$$

If the inverse system dynamics is asymptotically stable, then we say that R is strictly minimum phase. If, in addition, Ritself is asymptotically stable, then we say that R is strictly outer. In the linear case, (strictly) outer corresponds to all poles of the transfer function being in the open left-half plane and all zeros in the closed (open) left-half plane, while (strictly) minimum phase corresponds to all transmission zeros being in the closed (open) left-half plane.

# B. Reduction of (j, J)-Inner–Outer to j-Spectral Factorization: The Input–Output Level

The goal of this section is to obtain a (j, J)-inner-outer factorization for a system of the form of (1) under the assumption that the  $m \times m$  matrix function  $d^T(x)Jd(x)$  has a factorization

$$d^{T}(x)Jd(x) = \overline{d}^{T}(x)j\overline{d}(x) \tag{A2}$$

where  $x \to \overline{d}(x)$  is a smooth,  $m \times m$  matrix function with invertible values for all x. Note that in case p = m, j = J, and d(x) is invertible for all x, we can take simply  $\overline{d}(x) =$ d(x). If p > m and  $d(x)^T d(x)$  is invertible for all x with constant signature (i.e., the number of positive and negative eigenvalues) equal to the signature of j, then a Morse theory argument implies that such a factorization exists. We also impose in this section Assumption (A1) that  $x_0 = 0$  is an asymptotically stable equilibrium point for the uncontrolled system  $\dot{x} = a(x)$ .

To motivate our approach to the nonlinear (j, J)-inner-outer factorization problem, we recall the approach through *j*spectral factorization for the linear case. We start with a transfer function *G*. A property of (j, J)-inner functions is that  $\Theta(-s)^T J \Theta(s) = j$ . Hence, if *G* has a (j, J)-inner-outer factorization  $G = \Theta R$ , then

$$\begin{split} G(-s)^T J G(s) &= R(-s)^T \Theta(-s)^T J \Theta(s) R(s) \\ &= R(-s)^T j R(s). \end{split}$$

Conversely, if we produce an outer function R such that

$$G(-s)^T J G(s) = R(-s)^T j R(s)$$

and set  $\Theta = GR^{-1}$ , then  $\Theta$  has (j, J)-isometric values on the imaginary line. If G has a (j, J)-inner-outer factorization, then  $\Theta$  also has (j, J)-contractive values on the right-half plane and  $G = \Theta R$  is a (j, J)-inner-outer factorization for G. Our goal in this section is to show that a similar analysis holds in the nonlinear case at the input-output level; for the discrete-time case with j = J = I, see [3].

In this subsection we consider maps on  $L_2^m(\mathbf{R}^+)$  and ignore state-space representations. A map  $T_{\Theta}: L_2^m(\mathbf{R}^+) \to L_2^p(\mathbf{R}^+)$ is said to be input–output (j, J)-conservative if

$$\langle JT_{\Theta}(u), T_{\Theta}(u) \rangle_{L_2^p} = \langle ju, u \rangle_{L_2^m} \tag{7}$$

for all  $u \in L_2^m(\mathbb{R}^+)$ . We say that  $T_{\Theta}$  is input-output (j, J)-lossless (or (j, J)-lossless) if in addition

$$\langle JP_{\tau}T_{\Theta}(u), P_{\tau}T_{\Theta}(u)\rangle_{L_{2}^{m}} \leq \langle jP_{\tau}u, P_{\tau}u\rangle_{L_{2}^{p}}$$
(8)

for all  $u \in L_2^n$  and all real  $\tau$ . Here  $P_{\tau}$  is the truncation operator

$$\mathcal{P}_{\tau}f(t) = \begin{cases} f(t), & \text{for } t \leq \tau \\ 0, & \text{for } t > \tau. \end{cases}$$

Ì

This input-output notion of (j, J)-inner is closely related, but not equivalent, to the state-space notion introduced in Section II-A (see [35]).

A map  $T_R$  on  $L_2^m(\mathbf{R}^+)$  is input-output stable if  $T_R$  maps  $L_2^m(\mathbf{R}^+)$  into itself and is input-output outer if both  $T_R$  and its inverse  $T_R^{-1}$  are input-output stable. We assume throughout that all mappings on  $L_2(\mathbf{R}^+)$ -spaces are without bias in the sense that the image of zero is zero. In addition we say that the map  $T_R$  is causal if

$$P_{\tau} \circ T_R = P_{\tau} \circ T_R \circ P_{\tau}$$

for all real  $\tau$  and that  $T_R$  is bicausal if both  $T_R$  and  $(T_R)^{-1}$  are causal. We will discuss the distinction between internal stability and input-output stability in more detail in Section IV in connection with the  $H_{\infty}$ -control problem; for more information, see [28, Chapter 6].

Now let  $T_{\Sigma}$  be a mapping from  $L_2^m(\mathbf{R}^+)$  into  $L_2^p(\mathbf{R}^+)$ with  $T_{\Sigma}(0) = 0$ . The (j, J)-inner-outer factorization problem (at the input-output level) is to produce mappings  $T_{\Theta}$  and  $T_R$  which are (j, J)-inner and outer, respectively, in the input-output sense, such that  $T_{\Sigma} = T_{\Theta} \circ T_R$ . We assume that all maps are Frechet differentiable (see, for example, [31] for a discussion of the notion of Frechet derivative). For  $T_{\Sigma}$  a mapping as above, the Frechet derivative of  $T_{\Sigma}$ at the point  $u \in L_2^m(\mathbf{R}^+)$ , denoted by  $DT_{\Sigma}(u)$ , then is a linear mapping from  $L_2^m(\mathbf{R}^+)$  into  $L_2^p(\mathbf{R}^+)$  and hence has a transpose, denoted by  $[DT_{\Sigma}(u)]^T$ , with respect to the  $L_2$ inner products. The following result provides an extension of the connection between (j, J)-inner-outer factorization and *j*-spectral factorization to the nonlinear setting.

Theorem 1: Suppose that  $T_{\Sigma}: L_2^m(\mathbf{R}^+) \to L_2^p(\mathbf{R}^+)$  and  $T_R: L_2^m(\mathbf{R}^+) \to L_2^m(\mathbf{R}^+)$  are Frechet differentiable mappings without bias such that  $T_R$  is outer. Set  $T_{\Theta} = T_{\Sigma} \circ T_R^{-1}$ . Then  $T_{\Sigma} = T_{\Theta} \circ T_R^{-1}$  is a (j, J)-conservative-outer factorization of  $T_{\Sigma}$  if and only if

$$[DT_{\Sigma}(u)]^{T} \circ JT_{\Sigma}(u) = [DT_{R}(u)]^{T} \circ jT_{R}(u)$$
(9)

for all  $u \in L_2^m(\mathbf{R}^+)$ .

Moreover, if  $T_{\Sigma}$  has a (j, J)-inner-outer factorization (in the input-output sense) and if (9) holds for a certain bicausal outer  $T_R$ , then  $T_{\Theta} = T_{\Sigma} \circ T_R^{-1}$  is (j, J)-inner and  $T_{\Sigma} = T_{\Theta} \circ T_R$  is a (j, J)-inner-outer factorization of  $T_{\Sigma}$ .

The proof requires a preliminary fact of independent interest.

Proposition 1: Suppose  $T_{\Theta}$ :  $L_2^m(\mathbf{R}^+) \to L_2^p(\mathbf{R}^+)$  is (j, J)conservative in the input–output sense. Then

$$[DT_{\Theta}(u)]^T \circ JT_{\Theta}(u) = ju \tag{10}$$

for all  $u \in \mathbf{R}^m$ .

*Proof:* Differentiation of (7) with respect to u in the direction  $h \in L_2^m(\mathbb{R}^+)$  yields

$$\langle [DT_{\Theta}(u)]^T \circ JT_{\Theta}(u), h \rangle = \langle ju, h \rangle$$

from which (10) follows from the arbitrariness of h. Conversely, note that

$$\begin{aligned} \langle ju, u \rangle &- \langle JT_{\Theta}(u), T_{\Theta}(u) \rangle \\ &= \int_{0}^{1} \frac{d}{dt} \{ \langle j(tu), tu \rangle - \langle JT_{\Theta}(tu), T_{\Theta}(tu) \rangle \} dt \\ &= 2 \left( \int_{0}^{1} \{ \langle j(tu), u \rangle - \langle [DT_{\Theta}(tu)]^{T} \circ JT_{\Theta}(tu), u \rangle \} dt \right) \end{aligned}$$

whence (7) follows from (10).

Proof of Theorem 1: If  $T_{\Sigma} = T_{\Theta}T_R$  and  $T_{\Theta}$  is (j, J)conservative, then Proposition 1 (with  $T_R(u)$  in place of u)
gives us

$$[DT_{\Sigma}(u)]^{T} \circ JT_{\Sigma}(u)$$
  
=  $[DT_{R}(u)]^{T} \circ [DT_{\Theta}(T_{R}(u))]^{T} \circ JT_{\Theta}(T_{R}(u))$   
=  $[DT_{R}(u)]^{T} \circ jT_{R}(u)$ 

and (9) follows. Conversely, assume (9) and set  $T_{\Theta} = T_{\Sigma} \circ T_R^{-1}$ . Then

$$\begin{split} [DT_{\Theta}(u)]^{T} \circ JT_{\Theta}(u) \\ &= [DT_{R}(T_{R}^{-1}(u))]^{-T} \circ [DT_{\Sigma}(T_{R}^{-1}(u))]^{T} \\ &\circ JT_{\Sigma}(T_{R}^{-1}(u)) \\ &= [DT_{R}(T_{R}^{-1}(u))]^{-T} \circ [DT_{R}(T_{R}^{-1}(u))]^{T} \\ &\circ jT_{R}(T_{R}^{-1}(u)) \\ &= ju \end{split}$$

where the second line follows from (9) with  $T_R^{-1}(u)$  in place of u. From the other direction of Proposition 1, we conclude that  $\Theta$  is (j, J)-conservative.

Suppose now that  $T_{\Sigma}$  has a (j, J)-inner-outer factorization  $T_{\Sigma} = T_{\Theta'} \circ T_{R'}$  and that  $T_{\Sigma} = T_{\Theta} \circ T_R$  is a (j, J)-conservative-outer factorization as in the first part of the proof. Then we have the identity

$$T_{\Theta} = T_{\Theta'} \circ (T_{R'} \circ T_R^{-1}).$$

Since  $T_{R'} \circ T_R^{-1}$  is bicausal and outer, the (j, J)-inner property of  $T_{\Theta}$  carries over to the (j, J)-conservative  $T_{\Theta}$  as well. Hence the *j*-spectral factorization approach actually leads to a (j, J)-inner-outer factorization whenever such exists, as desired.

# C. Inner–Outer and Spectral Factorization for State-Space Systems

The goal of this section is to implement the *j*-spectral factorization problem (9) described in Theorem 1 (where  $T_{\Sigma}$  is known and  $T_R$  is to be found) in terms of state-space representations. We assume that  $T_{\Sigma}$  is the input-output map associated with a system  $\Sigma$  given by (1), and we seek an outer system R given by state-space equations of the form of (5) so that (9) holds. To achieve this we must discuss how to obtain a state-space realization for  $[DT_{\Sigma}]^T \circ JT_{\Sigma}$  and from this a realization for  $T_R$  so that (9) holds. The analysis parallels the approach to spectral factorization in [10].

To obtain a state-space realization of  $[DT_{\Sigma}]^T \circ JT_{\Sigma}$ , we first consider the Hamiltonian extension of  $\Sigma$  [where  $\Sigma$  is given by (1)] introduced in [12], namely

$$\dot{x} = a(x) + b(x)u$$

$$\dot{p} = -\left[\frac{\partial a}{\partial x}(x) + \frac{\partial b}{\partial x}(x)u\right]^{T} p - \frac{\partial^{T}c}{\partial x}(x)u_{a}$$

$$-u^{T}\frac{\partial^{T}d}{\partial x}(x)u_{a}, \quad u_{a} \in \mathbb{R}^{p}$$

$$y = c(x) + d(x)u$$

$$y_{a} = b^{T}(x)p + d^{T}(x)u_{a}, \quad y_{a} \in \mathbb{R}^{m}$$
(11)

which is a Hamiltonian system with state-space  $T^*M$  [the cotangent bundle of the state manifold M with local coordinates (x, p)] having inputs  $(u, u_a)$  and outputs  $(y, y_a)$ . Imposing the interconnection law  $u_a = Jy$  on (11) leads to the Hamiltonian system

$$[D\Sigma]^{T} \circ J\Sigma: \begin{cases} \dot{x} = \frac{\partial H}{\partial p} (x, p, u) \\ \dot{p} = -\frac{\partial H}{\partial x} (x, p, u) \\ y_{a} = \frac{\partial H}{\partial u} (x, p, u) \end{cases}$$
(12) with

with state-space  $T^*M$ , inputs  $u \in \mathbb{R}^m$ , outputs  $y_a \in \mathbb{R}^m$ , and the Hamiltonian function

$$H(x, p, u) = p^{T}(a(x) + b(x)u) + \frac{1}{2}(c(x) + d(x)u)^{T}$$
  
 
$$\cdot J(c(x) + d(x)u).$$
(13)

If we impose the initial condition (x, p) = (0, 0) and consider the input-output map  $T_{[D\Sigma]^T \circ J\Sigma}$  mapping u to  $y_a$ , it is not difficult to see that

$$T_{[D\Sigma]^T \circ J\Sigma}(u) = [DT_{\Sigma}(u)]^T \circ JT_{\Sigma}(u).$$

In particular, for a linear system  $\Sigma$  given by state-space equations

$$\dot{x} = Ax + Bu$$
$$y = Cx + Du$$

 $[D\Sigma]^T \circ J\Sigma$  is nothing other than the series interconnection of  $J \circ \Sigma$  and the adjoint system  $\Sigma^T$  having state-space equations

$$\dot{p} = -A^T p - C^T u_a$$
$$y_a = B^T p + D^T u_a$$

and has transfer matrix  $G^T(-s)JG(s)$ , where  $G(s) = C(sI - A)^{-j}B + D$ . The nonlinear case is somewhat more complicated. The action of the left factor  $[D\Sigma]^T$  in the composition  $[D\Sigma]^T \circ J\Sigma$  which produces the output  $y_a$  requires not only the output  $u_a = y$  of the right factor  $J\Sigma$  but also the state vector x of the right factor at each time t.

Our goal now is to produce an invertible outer system R as in (6) so that

$$[D\Sigma]^T \circ J\Sigma = [DR]^T \circ jR. \tag{14}$$

As we no longer insist on zero initial conditions for the state vectors in the state-space equations, this problem is actually somewhat more general than that discussed in Section II-B. Also, as was mentioned in Section II-B, the input-output notions of "(j, J)-inner" and of "outer" in general are not equivalent to the corresponding state-space notions. Here we shall implement the procedure outlined in Theorem 1 in terms of state-space representations. Rather than applying Theorem 1, we simply check directly for the "(j, J)-inner" and "outer" properties in the state-space sense of our solution. Thus Theorem 1 serves more as motivation rather than a logically integral part of the analysis.

If R is an outer system as in (5) (with state manifold denoted by  $\overline{M}$ ), then the same development as above gives that  $[DR]^T \circ jR$  has the form

$$\dot{\overline{x}} = \frac{\partial H_R}{\partial p} (\overline{x}, \overline{p}, u)$$
$$\dot{\overline{p}} = -\frac{\partial H_R}{\partial x} (\overline{x}, \overline{p}, u)$$
$$y_a = \frac{\partial H_R}{\partial u} (\overline{x}, \overline{p}, u)$$
(15)

where

$$H_{R}(\overline{x},\overline{p},u) = \overline{p}^{T}(\overline{a}(\overline{x}) + \overline{b}(\overline{x})u) + \frac{1}{2}(\overline{c}(\overline{x}) + \overline{d}(\overline{x})u)^{T}j(\overline{c}(\overline{x}) + \overline{d}(\overline{x})u).$$
(16)

We first show that the antistable invariant manifold of  $[DR]^T \circ jR$  for u = 0 (i.e., the set of points  $(\overline{x}, \overline{p})$  converging to  $(\overline{x}, \overline{p}) = (0,0)$  as  $t \to -\infty$  under the Hamiltonian flow) has the simple form

$$\{(\overline{x},\overline{p})\in T^*\overline{M}:\overline{x}=0\}.$$
 (AS)

Indeed, from (16) we see that

$$\frac{\partial H_R}{\partial p}(0,\overline{p},0) = \overline{p}^T \overline{a} (0) + \frac{1}{2} \overline{c}^T(0) j \overline{c}(0) = 0$$

for all  $\overline{p}$ . From (15) it follows that the manifold (AS) is invariant for (15) restricted to u = 0. Moreover, from (16) we have that

$$-\frac{\partial H_R}{\partial x}(0,\overline{p},0) = -\frac{\partial^T \overline{a}}{\partial x}(0)\overline{p}$$

By assumption  $x \to \overline{a}(x)$  is an asymptotically stable vector field, and hence the matrix  $-(\partial^T \overline{a}/\partial x)(0)$  is antistable (i.e., all its eigenvalues are in the closed right-half plane). Hence the Hamiltonian flow (15) (with u = 0) restricted to (AS) is antistable. By dimension count, it follows that (AS) is the antistable invariant manifold for (15) with u = 0.

Second, the inverse system is easily computed since by the assumption  $\overline{d}(x)$  is invertible; the result is

$$([DR]^{T} \circ jR)^{-1}: \begin{cases} \frac{\dot{x}}{x} = \frac{\partial H_{R}^{\times}}{\partial p} (\bar{x}, \bar{p}, y_{a}) \\ \frac{\dot{p}}{p} = -\frac{\partial H_{R}^{\times}}{\partial x} (\bar{x}, \bar{p}, y_{a}) \\ u = -\frac{\partial H_{R}^{\times}}{\partial y_{a}} (\bar{x}, \bar{p}, y_{a}) \end{cases}$$
(17)

with state-space  $T^*\overline{M}$ , inputs  $y_a$  and outputs u, where the inverse Hamiltonian  $H_R^{\times}(\overline{x}, \overline{p}, y_a)$  is obtained as the Legendre

## BALL AND VAN DER SCHAFT: J-INNER-OUTER FACTORIZATION

transform of  $H_R(\overline{x}, \overline{p}, u)$  with respect to u and  $y_a$ , i.e.,

$$H_{R}^{\times}(\overline{x},\overline{p},y_{a}) = H_{R}(\overline{x},\overline{p},u) - u^{T}y_{a},$$
  
$$u \text{ satisfying } \frac{\partial H_{R}}{\partial u}(\overline{x},\overline{p},u) = y_{a}.$$
(18)

(Note the somewhat unusual sign convention.) Direct computation using (16) gives

$$\begin{aligned} H_R^{\times}(\overline{x},\overline{p},y_a) &= \overline{p}^T(\overline{a}(\overline{x}) - \overline{b}(\overline{x})\overline{d}^{-1}(\overline{x})\overline{c}(\overline{x})) \\ &- \frac{1}{2}\overline{p}^T\overline{b}(\overline{x})\overline{d}^{-1}(\overline{x})j\overline{d}^{-T}(\overline{x})\overline{b}^T(\overline{x})\overline{p} \\ &+ (\overline{p}^T\overline{b}(\overline{x}) + \overline{c}^T(\overline{x})j\overline{d}(\overline{x}))\overline{d}^{-1}(\overline{x})j\overline{d}^{-T}(\overline{x})y_a \\ &- \frac{1}{2}y_a\overline{d}^{-1}(\overline{x})j\overline{d}^{-T}(\overline{x})y_a. \end{aligned}$$
(19)

We note from (19) that

$$-\frac{\partial H_R^{\times}}{\partial \overline{x}}(\overline{x},0,0)=0$$

From (17) we see that the other coordinate space

$$\{(\overline{x}, \overline{p}) \in T^* \overline{M} \colon \overline{p} = 0\}$$
(S)

is an invariant manifold of  $([DR]^T \circ jR)^{-1}$  with  $y_a = 0$ . Moreover, from (17) and (19) we see that the restriction of the dynamics of  $([DR]^T \circ jR)^{-1}$  with  $y_a = 0$  to this manifold is given by the vector field

$$\frac{\partial H_R^{\times}}{\partial \overline{p}}(\overline{x}, 0, 0) = \overline{a}^{\times}(\overline{x})$$

where

$$\overline{a}^{\times}(\overline{x}) := \overline{a}(\overline{x}) - \overline{b}(\overline{x})\overline{d}^{-1}(\overline{x})\overline{c}(\overline{x}).$$

If R is (strictly) outer, then  $\overline{a}^{\times}(\overline{x})$  is Lyapunov (asymptotically) stable. By dimension count we conclude that the manifold (S) is the stable invariant manifold for the Hamiltonian flow associated with the inverse system  $([DR]^T \circ jR)^{-1}$  with  $y_a = 0$ . To factor  $[D\Sigma]^T \circ J\Sigma$  as  $[DR]^T \circ jR$ , the idea now is to find a canonical change of coordinates  $(x, p) \to (\overline{x}, \overline{p})$  so that the system  $[D\Sigma]^T \circ J\Sigma$  given by (12) and (13) expressed in the new coordinates will have the form of (15) and (16); from (16) we can then read off the desired coefficients  $\overline{a}(\overline{x}), \overline{b}(\overline{x}), \overline{c}(\overline{x}), \overline{d}(\overline{x})$  for the desired outer factor. It is crucial that the change of coordinates be canonical to preserve the Hamiltonian structure (see e.g., [1]).

First note from Assumption (A2) the fact that  $d^T(x)Jd(x)$ is invertible guarantees the causal invertibility of the system  $[D\Sigma]^T \circ J\Sigma$ . Indeed, analogously to the procedure for  $[DR]^T \circ jR$ , the inverse system  $([D\Sigma]^T \circ J\Sigma)^{-1}$  is the Hamiltonian system

$$([D\Sigma]^{T} \circ J\Sigma)^{-1} \begin{cases} \dot{x} = \frac{\partial H^{\times}}{\partial p} (x, p, y_{a}) \\ \dot{p} = -\frac{\partial H^{\times}}{\partial x} (x, p, y_{a}) \\ u = -\frac{\partial H^{\times}}{\partial y_{a}} (x, p, y_{a}) \end{cases}$$
(20)

state-space  $T^*M$ , inputs  $y_a$ , and outputs u, where the inverse Hamiltonian  $H^{\times}$  is the Legendre transform of H(x, p, u) with

respect to u and  $y_a$ , i.e.,

$$H^{\times}(x, p, y_a) = H(x, p, u) - u^T y_a$$
  
u satisfying  $\frac{\partial H}{\partial u}(x, p, u) = y_a.$  (21)

Explicitly, this works out to be

$$H^{\times}(x, p, y_{a}) = p^{T}(a(x) - b(x)E^{-1}(x)d^{T}(x)Jc(x)) + \frac{1}{2}c^{T}(x)(J - Jd(x)E^{-1}(x)d^{T}(x)J)c(x) - \frac{1}{2}p^{T}b(x)E^{-1}(x)b^{T}(x)p + (p^{T}b(x) + c^{T}(x)Jd(x))E^{-1}(x)y_{a} - \frac{1}{2}y_{a}^{T}E^{-1}(x)y_{a}$$
(22)

where  $E(x) = d^T(x)Jd(x)$ .

The desired canonical change of coordinates  $(x, p) \rightarrow (\overline{x}, \overline{p})$ must be such that the antistable invariant manifold for the Hamiltonian flow induced by H (with u = 0) is given by  $\{(\overline{x}, \overline{p}) \in T^*M: \overline{x} = 0\}$  and is such that the manifold  $\{(\overline{x}, \overline{p}) \in T^*M: \overline{p} = 0\}$  is a Lyapunov (or asymptotically, in the strict outer case) stable invariant manifold of  $H^{\times}$  (with  $y_a = 0$ ). Note next that by the same calculation as was done above for  $H_R$ , our standing Assumption (A1) that ais asymptotically stable implies that the antistable manifold for H (with u = 0) is  $\{(x, p) \in T^*M: x = 0\}$ . To compute a Lagrangian invariant manifold for  $([D\Sigma]^T \circ J\Sigma)^{-1}$ with  $y_a = 0$ , we consider the Hamilton–Jacobi equation  $H^{\times}(x, P_x^T(x), 0) = 0$  for some smooth function P(x) with P(0) = 0, i.e.,

$$P_{x}(x)[a(x) - b(x)E^{-1}(x)d^{T}(x)Jc(x)] + \frac{1}{2}c^{T}(x)[J - Jd(x)E^{-1}(x)d^{T}(x)J]c(x) - \frac{1}{2}P_{x}(x)b(x)E^{-1}(x)b^{T}(x)P_{x}^{T}(x) = 0, \qquad P(0) = 0$$
(23)

with the stability side condition

$$\frac{\partial H^{\times}}{\partial p}(x, P_x^T(x), 0)$$

$$= a(x) - b(x)E^{-1}(x)[d^T(x)Jc(x) + b^T(x)P_x^T(x)]$$
is Lyapunov stable. (24)

Suppose that there exists a smooth solution P to (23) and (24). Then it is well known (see [22]) that the submanifold of  $T^*M$  given by

$$\{(x, p): p = P_x^T(x)\}$$
(25)

is an invariant manifold for  $([D\Sigma]^T \circ J\Sigma)^{-1}$  with  $y_a = 0$ , on which the dynamics coordinated by x are given by the Lyapunov stable vector field given in (24). This leads to the canonical transformation  $(x, p) \to (x, \overline{p})$  with

$$p = \overline{p} + P_x^T(x).$$

Clearly, in these new coordinates,  $\{(x, \overline{p}): x = 0\}$  remains the antistable invariant manifold for  $[D\Sigma]^T \circ J\Sigma$  with u = 0, and the invariant manifold (25) for  $([D\Sigma]^T \circ J\Sigma)^{-1}$  with

 $y_a = 0$  is given simply as  $\{(x, \overline{p}): \overline{p} = 0\}$ . The next step is to compute the Hamiltonian H(x, p, u) in these new coordinates

$$\overline{H}(x,\overline{p},u) = H(x,\overline{p} + P_x^T(x), u)$$

$$= \overline{p}^T(a(x) + b(x)u) + \frac{1}{2}(\overline{c}(x) + \overline{d}(x)u)^T j(\overline{c}(x) + \overline{d}(x)u)$$
(26)

where  $\overline{d}(x)$  is chosen as in (A2)

$$d^{T}(x)Jd(x) = \overline{d}^{T}(x)j\overline{d}(x), \qquad \overline{d}(x) \text{ of size } m \times m$$
 (27)

and

$$\overline{c}(x) = \overline{d}(x)E^{-1}(x)[d^T(x)Jc(x) + b^T(x)P_x^T(x)].$$
(28)

Comparison of (26) and (16) now gives that  $[D\Sigma]^T \circ J\Sigma = [DR]^T \circ jR$  where R is the new system defined by

$$R: \begin{cases} \dot{x} = a(x) + b(x)u\\ \overline{y} = \overline{c}(x) + \overline{d}(x)u \end{cases}$$
(29)

with  $\overline{c}$  and  $\overline{d}$  as in (28) and (27), respectively. Clearly, from (A1) R is asymptotically stable. The inverse dynamics for R is given by

$$a(x) - b(x)\overline{d}^{-1}(x)\overline{c}(x) = a(x) - b(x)E^{-1}(x)[d^{T}(x)Jc(x) + b^{T}(x)P_{x}^{T}(x)]$$

which is Lyapunov stable by (24). Thus R is also minimum phase and so is an outer factor of  $\Sigma$ .

The next step is to compute  $\Theta$  as  $\Theta = \Sigma \circ R^{-1}$ . From (29) we see that  $R^{-1}$  is given by

$$R^{-1}: \begin{cases} \dot{\tilde{x}} = a(\tilde{x}) - b(\tilde{x})\overline{d}^{-1}(\tilde{x})\overline{c}(\tilde{x}) + b(\tilde{x})\overline{d}^{-1}(\tilde{x})\overline{y} \\ u = -\overline{d}^{-1}(\tilde{x})\overline{c}(\tilde{x}) + \overline{d}(\tilde{x})^{-1}\overline{y}. \end{cases}$$
(30)

Combine with (1) for  $\Sigma$  and use the interconnection law that u in (1) is equal to u in (30) to get

$$\Sigma \circ R^{-1}: \begin{cases} \dot{x} = a(x) - b(x)\overline{d}^{-1}(x)\overline{c}(\tilde{x}) + b(x)\overline{d}^{-1}(\tilde{x})\overline{y} \\ \dot{x} = a(\tilde{x}) - b(\tilde{x})\overline{d}^{-1}(\tilde{x})\overline{c}(\tilde{x}) + b(\tilde{x})\overline{d}^{-1}(\tilde{x})\overline{y} \\ y = c(x) - d(x)\overline{d}^{-1}(\tilde{x})\overline{c}(\tilde{x}) + d(x)\overline{d}(\tilde{x})^{-1}\overline{y}. \end{cases}$$

$$(31)$$

Note that if the initial state  $(x(0), \tilde{x}(0))$  is on the diagonal  $x(0) = \tilde{x}(0)$ , then the state vector remains on the diagonal for all t > 0  $(x(t) = \tilde{x}(t))$ , and the reduced system  $\Theta$  (with state vector (x, x) parameterized simply by x) is given by

$$\Theta: \begin{cases} \dot{x} = a(x) - b(x)\overline{d}^{-1}(x)\overline{c}(x) + b(x)\overline{d}^{-1}(x)\overline{y} \\ y = c(x) - d(x)\overline{d}^{-1}(x)\overline{c}(x) + d(x)\overline{d}^{-1}(x)\overline{y}. \end{cases}$$
(32)

It is not difficult to check, by using (23) satisfied by P together with (27) and (28) for  $\overline{d}$  and  $\overline{c}$ , that the system of equations

$$P_{x}(x)[a(x) - b(x)\overline{d}^{-1}(x)\overline{c}(x)] + \frac{1}{2}[c(x) - d(x)\overline{d}^{-1}(x)\overline{c}(x)]^{T} \\ \cdot J[c(x) - d(x)\overline{d}^{-1}(x)\overline{c}(x)] = 0 \\ P_{x}(x)b(x)\overline{d}^{-1}(x) + [c(x) - d(x)\overline{d}^{-1}(x)\overline{c}(x)]^{T} \\ \cdot Jd(x)\overline{d}^{-1}(x) = 0 \\ [d(x)\overline{d}^{-1}(x)]^{T}J[d(x)\overline{d}^{-1}(x)] = j$$
(33)

holds for all x, and hence P is a storage function for  $\Theta$  with supply rate  $\frac{1}{2}\overline{y}^T j\overline{y} - \frac{1}{2}y^T Jy$  if  $P(x) \ge 0$ . One also easily

checks that  $\Sigma = \Theta \circ R$  in the sense that given any initial state  $x_0$  for  $\Sigma$ , there is a choice of initial state for  $\Theta \circ R$  (namely,  $(x_0, x_0)$ ) so that the input-output map for  $\Sigma$  is the same as the input-output map associated with  $\Theta \circ R$ . The following summarizes our discussion.

Theorem 2: Assume that (A1) and (A2) hold. Suppose that there exists a solution  $P \ge 0$  to (23) and (24). Then a (j, J)-inner-outer factorization of  $\Sigma$  is  $\Theta \circ R$  with R and  $\Theta$  defined by (29) and (32), respectively.

*Remark:* In the context of  $H_{\infty}$  control, it is important to consider the larger system  $\Sigma \circ R^{-1}$  defined by (31) rather than  $\Theta$ . While it is clear that  $\Sigma = (\Sigma \circ R^{-1}) \circ R$  at the input-output level, there is no obvious storage function for  $\Sigma \circ R^{-1}$ , and hence it is not clear if  $\Sigma \circ R^{-1}$  is (j, J)-inner. We shall return to this topic in Section IV.

Let us now consider the case where the inverse system  $([D\Sigma]^T \circ J\Sigma)^{-1}$  for  $y_a = 0$  does not possess any dynamics corresponding to purely imaginary eigenvalues, that is, where the linearization of  $([D\Sigma]^T \circ J\Sigma)$  does not have purely imaginary transmission zeros. In this situation the stable invariant manifold for the Hamiltonian flow induced by  $H^{\times}$  (with  $y_a = 0$ ) is automatically Lagrangian (see [22]), and one can state a local version of Theorem 2 as follows. Consider the linearization at (0, 0) of (21) for  $y_a = 0$ . This gives rise to a linear Hamiltonian system

$$\begin{bmatrix} \dot{x} \\ \dot{p} \end{bmatrix} = \begin{bmatrix} A & -R \\ -Q & -A^T \end{bmatrix} \begin{bmatrix} x \\ p \end{bmatrix}$$

$$A = \frac{\partial a}{\partial x}(0) - b(0)E^{-1}(0)d^T(0)\frac{\partial c}{\partial x}(0)$$

$$R = b(0)E^{-1}(0)b^T(0)$$

$$Q = \frac{\partial^T c}{\partial x}(0)[J - Jd(0)E^{-1}(0)d^T(0)J]\frac{\partial^T c}{\partial x}(0).$$
(34)

Assume also that

- i) The Hamiltonian matrix in (34) does not have purely imaginary eigenvalues; and
- ii) The generalized eigenspace corresponding to eigenvalues in the open left-half complex plane for the Hamiltonian matrix in (34) has the form span  $\begin{bmatrix} I \\ X \end{bmatrix}$  for some positive semidefinite matrix X.

Condition ii) is equivalent to: there exists a positive semidefinite solution  $X \ge 0$  to the algebraic Riccati equation

$$XA + A^T X - XRX + Q = 0 \tag{35}$$

with stability side condition

$$A - RX$$
 is asymptotically stable. (36)

(Here A, R, Q are as in (34).) Then we have the following result. For guidelines to the details of the proof, we refer to [23].

Theorem 3: Assume (A1) and (A2) hold. Assume in addition that conditions i) and ii) above are satisfied. Then there exists a neighborhood  $W \subset M$  of  $x_0 = 0$  and a smooth function  $P: W \to \mathbf{R}$  with  $P(x) \ge 0$  for  $x \in W$  which is the unique solution of (23) together with the strengthened local

#### BALL AND VAN DER SCHAFT: J-INNER-OUTER FACTORIZATION

version of (24)

$$a(x) - b(x)E^{-1}(x)[d^{T}(x)Jc(x) + b^{T}(x)P_{x}^{T}(x)]$$
  
is asymptotically stable for  $x \in W$ . (37)

If we define R and  $\Theta$  as in (29) and (32), respectively, for  $x \in W$ , then  $\Sigma = \Theta \circ R$  is a (j, J)-inner-outer factorization of  $\Sigma$  on W, where the outer factor R is strictly minimum phase on W.

Furthermore, if the function  $P: W \to \mathbf{R}$  can be extended to a global solution  $P: M \to \mathbf{R}$  with  $a(x) - b(x)E^{-1}(x)[d^T(x)Jc(x) + b^T(x)P_x^T(x)]$  globally asymptotically stable and with the nonnegativity constraint  $P(x) \ge 0$  holding for all  $x \in M$ , then  $\Theta$  and R are globally defined inner and outer factors with R strictly minimum phase on M.

*Remark:* In the case where  $J = I_p$  and  $j = I_m$ , one can use inertia-type theorems to guarantee a unique local solution of (23) and (24). We will discuss this case in more detail in [27].

#### III. APPLICATION TO $L_2$ -GAIN

As an application of the theory of nonlinear spectral factorization developed in Section II, we give here a new interpretation of the state-space characterization of systems having finite gain at most equal to a tolerance level  $\gamma$ .

Let us consider as in Section II a smooth nonlinear statespace system  $\Sigma$  which is affine in the inputs

$$\Sigma: \begin{cases} \dot{x} = a(x) + b(x)u, & u \in \mathbf{R}^m \\ y = c(x) + d(x)u, & y \in \mathbf{R}^p. \end{cases}$$
(38)

We assume that

$$a(x)$$
 is asymptotically stable (A1)

and that

$$\gamma^2 I_m - d^T(x)d(x) = \overline{d}^T(x)\overline{d}(x) \tag{A2}$$

for a smooth, everywhere invertible  $m \times m$  matrix function  $\overline{d}(x)$ , where  $\gamma$  is a fixed positive number. As usual, we also assume that a(0) = 0, c(0) = 0. Assume first that the associated input-output map  $T_{\Sigma}: u \to y$  (defined with the state initialized to the equilibrium point x(0) = 0) is a well-defined mapping from  $L_{2,e}^m(\mathbf{R}^+)$  into  $L_{2,e}^p(\mathbf{R}^+)$  (where  $L_{2,e}^n(\mathbf{R}^+)$ , in general, consists of  $\mathbf{R}^n$ -vector functions defined on  $\mathbf{R}^+ = [0, \infty)$  which are square-integrable in norm over each finite interval). Then we say that  $T_{\Sigma}$  has finite gain less than or equal to  $\gamma$  if

$$||T_{\Sigma}(u)||_{2} \leq \gamma ||u||_{2}, \qquad u \in L_{2}^{m}(\mathbf{R}^{+}).$$
(39)

By the causality of  $T_{\Sigma}$ , this in turn is equivalent to

$$\int_0^{t_1} \|y(t)\|^2 dt \le \gamma^2 \int_0^{t_1} \|u(t)\|^2 dt$$

for all  $t_1 > 0$  whenever  $y = T_{\Sigma}(u)$ .

A closely related state-space idea is that there exists a storage function V(x) for  $\Sigma$  with respect to the supply rate  $\frac{1}{2}\gamma^2 ||u||^2 - \frac{1}{2}||y||^2$  (in the sense of dissipative systems rather

than of lossless systems as in Section II—see [29]), i.e., that there exists a function  $V(x) \ge 0$  on M so that

$$V(x(t_1)) - V(x(t_0)) \le \frac{1}{2} \int_{t_0}^{t_1} (\gamma^2 ||u(t)||^2 - ||y(t)||^2) dt$$
$$V(0) = 0$$
(40)

for all  $t_0 \leq t_1$  and  $u(\cdot)$  with  $x(t_1)$  denoting the state at time  $t_1$  resulting from initial state  $x(t_0)$  at time  $t_0$  and input  $u(\cdot)$  on the time interval  $[t_0, t_1]$ . In particular, if we take  $t_0 = 0, x(t_0) = 0$ , we recover (39) as a consequence of (40). If the system is reachable, conversely one can construct a function  $V(x) \geq 0$  satisfying (40) as a consequence of (39). If V is differentiable, then (40) can be equivalently expressed as

$$V_x(x)[a(x) + b(x)u] + \frac{1}{2}[c(x) + d(x)u]^T[c(x) + d(x)u] - \frac{1}{2}\gamma^2 u^T u \le 0, \qquad V(0) = 0$$
(41)

for all x and u. As the expression on the left is quadratic in u, we can compute its maximum over u explicitly. Then (41) has an equivalent formulation

$$V_{x}(x)[a(x) + b(x)(\gamma^{2}I_{m} - d^{T}(x)d(x))^{-1}d^{T}(x)c(x)] + \frac{1}{2}c^{T}(x)(I_{p} - \gamma^{-2}d(x)d^{T}(x))^{-1}c(x) + \frac{1}{2}V_{x}(x)b(x)(\gamma^{2}I_{m} - d^{T}(x)d(x))^{-1}b^{T}(x)V_{x}(x) \leq 0 V(0) = 0$$

$$(42)$$

for all x.

An alternative derivation of (41) from the point of view of spectral factorization can be done as follows. An obvious sufficient condition for (39) to hold is that there exists a smooth system R

$$R: \begin{cases} \dot{x} = \overline{a}(x) + \overline{b}(x)u\\ \overline{y} = \overline{c}(x) + \overline{d}(x)u \end{cases}$$
(43)

such that

$$\frac{1}{2}(\gamma^2 ||u||^2 - ||T_{\Sigma}(u)||^2) = \frac{1}{2}||T_R(u)||^2$$
(44)

for all u in  $L_2^m(\mathbf{R}^+)$ . Frechet-differentiation of (44) with respect to u then gives

$$\gamma^2 u - [DT_{\Sigma}(u)]^T \circ T_{\Sigma}(u) = [DT_R(u)]^T \circ T_R(u).$$
 (45)

Conversely, (45), together with initial conditions  $T_R(0) = 0, T_{\Sigma}(0) = 0$ , implies (44) by a straightforward integration along the line segment  $\{tu: 0 \le t \le 1\} \subset L_2^m(\mathbf{R}^+)$ .

Thus to show that  $T_{\Sigma}$  has gain at most  $\gamma$ , it suffices to produce a system R so that (45) is satisfied. But a state-space realization of the map  $u \rightarrow \gamma^2 u - [DT_{\Sigma}(u)]^T \circ T_{\Sigma}(u)$ , computed in the form  $u \rightarrow \gamma^2 u + [DT_{\Sigma}(u)]^T \circ (-T_{\Sigma}(u))$ , is

$$\gamma^{2} I_{m} - [D\Sigma]^{T} \circ \Sigma \begin{cases} \dot{x} = \frac{\partial \tilde{H}}{\partial p} (x, p, u) \\ \dot{p} = -\frac{\partial \tilde{H}}{\partial x} (x, p, u) \\ y_{a} = \frac{\partial \tilde{H}}{\partial u} (x, p, u) \end{cases}$$
(46)

IEEE TRANSACTIONS ON AUTOMATIC CONTROL, VOL. 41, NO. 3, MARCH 1996

with state-space  $T^*M$ , inputs  $u \in \mathbb{R}^m$ , and outputs  $y_a \in \mathbb{R}^m$ where the Hamiltonian function  $\tilde{H}(x, p, u)$  is given by

$$\tilde{H}(x, p, u) = p^{T}(a(x) + b(x)u) - \frac{1}{2}(c(x) + d(x)u)^{T}$$
$$(c(x) + d(x)u) + \frac{1}{2}\gamma^{2}u^{T}u.$$
(47)

(This amounts to the Hamiltonian extension (11) of  $\Sigma$  with connection law  $u_a = -y$  in series connection with the memoryless system  $u \rightarrow \gamma^2 u$ .) Due to Assumption (A2), the inverse system is easily computed; it is again a Hamiltonian system

$$(\gamma^{2}I_{m} - [D\Sigma]^{T} \circ \Sigma)^{-1} \begin{cases} \dot{x} = \frac{\partial H^{\times}}{\partial p} (x, p, y_{a}) \\ \dot{p} = -\frac{\partial \tilde{H}^{\times}}{\partial x} (x, p, y_{a}) \\ u = -\frac{\partial \tilde{H}^{\times}}{\partial y_{a}} (x, p, y_{a}) \end{cases}$$
(48)

with the Hamiltonian equal to the Legendre transform of H(x, p, u) with respect to u and  $y_a$ 

$$\tilde{H}^{\times}(x, p, y_a) = \tilde{H}(x, p, u) - u^T y_a,$$
  
$$u \text{ satisfying } y_a = \frac{\partial \tilde{H}}{\partial u}(x, p, u).$$
(49)

From (47),  $\tilde{H}^{\times}(x, p, y_a)$  works out explicitly to be

$$\begin{split} \tilde{H}^{\times}(x,p,y_{a}) &= p^{T}[a(x) + b(x)(\gamma^{2} - d^{T}(x)d(x))^{-1} \\ &\cdot d^{T}(x)c(x)] - \frac{1}{2}p^{T}b(x) \\ &\cdot (\gamma^{2}I_{m} - d^{T}(x)d(x))^{-1}b^{T}(x)p \\ &- \frac{1}{2}c^{T}(x)(I_{p} - \gamma^{-2}d(x)d^{T}(x))^{-1}c(x) \\ &+ p^{T}b(x)(\gamma^{2}I_{m} - d^{T}(x)d(x))^{-1}y_{a} \\ &+ c^{T}(x)d(x)(\gamma^{2}I_{m} - d^{T}(x)d(x))^{-1}y_{a} \\ &- \frac{1}{2}y_{a}^{T}(\gamma^{2}I_{m} - d^{T}(x)d(x))^{-1}y_{a}. \end{split}$$
(50)

To factor  $\gamma^2 I_m - [D\Sigma]^T \circ \Sigma$  as  $[DR]^T \circ R$  with R stable, we know from experience in Section II that we need to compute the antistable invariant manifold of  $\gamma^2 I_m$  –  $[D\Sigma]^T \circ \overline{\Sigma}$  with u = 0 and some invariant manifold of  $(\gamma^2 I_m - [D\Sigma]^T \circ \Sigma)^{-1}$  with  $y_a = 0$ . The former is simply  $\{(x, p): x = 0\}$  due to Assumption (A1). As for the latter, we consider the Hamilton-Jacobi equation

$$\tilde{H}^{\times}(x, -V_x^T(x), 0) = 0, \qquad V(0) = 0.$$
 (51)

This is the same as (42) but with an equality rather than inequality. We introduce the change of coordinates  $p = \overline{p}$  –  $V_x^T(x)$  and compute the original Hamiltonian in terms of these new coordinates. The result is

*m* . . .

$$\overline{\widetilde{H}}(x,\overline{p},u) = \overline{p}^{T}[a(x) + b(x)u] - \frac{1}{2}[c(x) + d(x)u]^{T} \\
\cdot [c(x) + d(x)u] + \frac{1}{2}\gamma^{2}u^{T}u - V_{x}(x) \\
\cdot [a(x) + b(x)u] \\
= \overline{p}^{T}[a(x) + b(x)u] + \frac{1}{2}[\overline{c}(x) + \overline{d}(x)u]^{T} \\
\cdot [\overline{c}(x) + \overline{d}(x)u]$$
(52)

$$\overline{l}^T(x)\overline{d}(x) = \gamma^2 I_m - d^T(x)d(x)$$
(53)

as in (A2), and

$$\overline{c}(x) = -\overline{d}^{-T}(x)(d^T(x)c(x) + b^T(x)V_x^T(x))$$
(54)

where we used (51). By comparison with (19) (where now  $j = I_m$ ), we see that  $\gamma^2 I_m - [D\Sigma]^T \circ \Sigma = [DR]^T \circ R$ with R given by

$$R: \begin{cases} \dot{x} = a(x) + b(x)u\\ \overline{y} = \overline{c}(x) + \overline{d}(x)u \end{cases}$$
(55)

with  $\overline{c}$  and  $\overline{d}$  as in (54) and (53), respectively. We summarize the discussion as follows. The result is an alternate interpretation for the role of (42) for a system having gain  $< \gamma$ . One can also verify the result directly by using a standard completion-of-squares argument based on (51).

*Theorem 4*: Suppose that  $\Sigma$  is a system satisfying Assumptions (A1) and (A2), and suppose that  $V(x) \ge 0$  is a smooth solution of (42) with equality. Then the input–output map  $T_{\Sigma}$ of  $\Sigma$  (defined with x(0) = 0) satisfies

$$\gamma^2 ||u||^2 - ||T_{\Sigma}(u)||^2 = ||T_R(u)||^2$$

where R is (55) with  $\overline{c}(x)$  and  $\overline{d}(x)$  given by (53) and (54), and hence  $T_{\Sigma}$  has finite gain at most  $\gamma$ .

# IV. Nonlinear $H_{\infty}$ -Control

Consider a nonlinear system  $P: \begin{bmatrix} w \\ u \end{bmatrix} \rightarrow \begin{bmatrix} z \\ y \end{bmatrix}$  with statespace equations

$$P: \begin{cases} \dot{x} = A(x) + B_1(x)w + B_2(x)u \\ z = C_1(x) + D_{12}(x)u \\ y = C_2(x) + D_{21}(x)w. \end{cases}$$
(56)

Here w is a reference and/or disturbance signal, u is the control signal, z is an error signal, and y is a measurement signal with values in  $\mathbb{R}^{n_w}, \mathbb{R}^{n_u}, \mathbb{R}^{n_z}$ , and  $\mathbb{R}^{n_y}$ , respectively. The  $H_{\infty}$ problem is to design a dynamic compensator  $K: y \to u$  with state-space equations

$$\dot{x}_K = A_K(x_K) + B_K(x_K)y$$
$$u = C_K(x_K) + D_K(x_K)y$$
(57)

so that

- i) System (56) and (57) is "internally stable," and
- ii) The closed-loop input-output map (with x(0) = $0, x_K(0) = 0$  has  $L_2$ -gain at most  $\gamma$

$$||z||_2 \leq \gamma ||w||_2$$

for all  $w \in L_2^{n_w}(\mathbf{R}^+)$ .

Here we assume that A(0) = 0,  $C_1(0) = 0$ ,  $C_2(0) = 0$ ,  $A_K(0) = 0, C_K(0) = 0$  and all functions  $A, B_1, B_2, C_1, D_{12}, D_{12}$  $C_2, D_{21}$  are smooth.

The precise notion of "internally stable" is taken in two distinct senses (one input-output, the other internal state space) in the literature. These two senses are equivalent in the linear case but not, in general, in the nonlinear case. For the input-output sense of internal stability, we say that (56) and (57) are internally stable (IO sense) if the enlarged system of equations

$$\dot{x} = A(x) + B_1(x)w + B_2(x)u$$

$$z = C_1(x) + D_{12}(x)u$$

$$y = C_2(x) + D_{21}(x)w$$

$$\dot{x}_K = A_K(x_K) + B_K(x_K)(y + v_2)$$

$$u = v_1 + C_K(x_K) + D_K(x_K)(y + v_2)$$
(58)

with both x(0) = 0 and  $x_K(0) = 0$  as initial conditions, determines a well-defined causal map

$$H: \begin{bmatrix} w \\ v_1 \\ v_2 \end{bmatrix} \to \begin{bmatrix} z \\ u \\ y \end{bmatrix}$$

which is stable with finite gain. On the other hand, (56) and (57) are said to be internally stable (internal state-space sense) if the vector field

$$(x, x_K) \to (A(x) + B_2(x)C_K(x_K) + B_2(x)D_K(x_K)C_2(x),$$
  
 $A_K(x) + B_K(x_K)C_2(x)))$ 

associated with the dynamics of the closed-loop system, (56) and (57), is asymptotically stable in the sense that  $(x(t), x_K(t)) \rightarrow (0, 0)$  for any choice of initialization  $(x(0), x_K(0))$ , where  $(x(t), x_K(t))$  is determined by (56) and (57) with w = 0. With extra hypotheses, it is possible to prove input-output stability from internal stability in the state-space sense and conversely. More precisely, if the closed-loop system is globally reachable and uniformly observable, then internal asymptotic stability (state-space sense) follows from input-output stability with finite gain (see [32]); conversely, if the mappings  $A(x), B_1(x), B_2(x), C_1(x), C_2(x), D_{12}(x), D_{21}(x), A_K(x_K),$  $B_K(x_K), C_K(x)$  all satisfy a global Lipschitz estimate and (0, 0) is a global, exponentially stable equilibrium point for the closed-loop system, then the closed-loop system is stable with finite gain (see [34]). A simple example of a system which is internally stable (state-space sense) but does not have finite  $L_2$ -gain is

$$\dot{x} = -x^3 + u$$
$$u = x.$$

A number of recent papers (see [5], [8], [11], [16], [17], [20], and [22]-[24]) have shown how a solution of the nonlinear  $H_{\infty}$ -control problem can be obtained from a smooth solution of a Hamilton-Jacobi equation for the state-feedback case (see [5], [11], [22], and [23]) or (at least locally) from smooth solutions of a coupled pair of Hamilton-Jacobi equations for the measurement feedback case (see [8], [16], [17], [20], and [24]). Our purpose here is to recover this type of result via a (j, J)-inner-outer factorization procedure for the disturbance feedforward case (to be described below).

Additional standard assumptions (called the regular case in the linear theory) are

$$D_{12}^T(x)D_{12}(x) > 0$$
 for all  $x$  (A3)

and

$$D_{21}(x)D_{21}^T(x) > 0$$
 for all  $x$ . (A4)

For the discussion here we shall assume the strengthened form of (A4)

$$D_{21}(x)$$
 is square and invertible for all  $x$  (A4')

(and hence  $n_y = n_w$ ). In this case we can solve the last of (56) for the disturbance w in terms of the measurement y. With this assumption in force, (56) can be rearranged to form a system G having the same trajectories as P but with inputs equal to (u, y) and outputs equal to (z, w)

$$G: \begin{cases} \dot{x} = [A(x) - B_1(x)D_{21}^{-1}(x)C_2(x)] \\ + B_2(x)u + B_1(x)D_{21}^{-1}(x)y \\ z = C_1(x) + D_{12}(x)u \\ \tilde{w} = -\gamma D_{21}(x)^{-1}C_2(x) + \gamma D_{21}(x)^{-1}y \end{cases}$$
(59)

where we have set  $\tilde{w} = \gamma w$ . We shall assume in addition that the vector field  $a(x) := A(x) - B_1(x)D_{21}^{-1}(x)C_2(x)$  is asymptotically stable. In this case, the original problem for Pis in the disturbance feedforward Case in the classification of [20].

The article [9] relates solutions K of the nonlinear  $H_{\infty}$ control problem for the system P in (56) (with internal stability taken in the input-output sense and where one seeks gain at most equal to  $\gamma$ ) to (j, J)-inner-outer factorization  $G = \Theta \circ R$  of the system G in (59)  $\left( \text{with } J = \begin{bmatrix} I_{n_z} & 0\\ 0 & -I_{n_w} \end{bmatrix} \right)$ and  $j = \begin{bmatrix} I_{n_u} & 0 \\ 0 & -I_{n_y} \end{bmatrix}$ . For the results to be valid, R is required to be incrementally outer, i.e., the input-output map

 $T_R$  associated with R (with zero initial condition on the state vector) is incrementally stable in the sense that

$$\|P_{\tau}[T_R(u+v) - T_R(u)]\| \le \tilde{\gamma} \|P_{\tau}v\|)$$

for some  $\tilde{\gamma} < \infty$ , for all  $\tau < \infty$  and  $u \in L_{2,e}(\mathbf{R}^+), v \in$  $L_2(\mathbf{R}^+)$ , and similarly for  $(T_R)^{-1}$ . Then under appropriate conditions one can use  $(T_R)^{-1}$  to parameterize many solutions K of the nonlinear  $H_{\infty}$ -problem (see [9]). As the present paper is primarily concerned with state-space representations, we do not use these results in the derivation to follow but, rather, analyze directly the validity of the same procedure for giving solutions in the input-output sense.

Our aim here is to use the results of Section II to compute explicitly a state-space realization of a (j, J)-inner-outer factorization of G as in (59) in terms of a state-space realization of the original plant P and to give conditions for compensators K derived in a simple way from the outer factor R to solve the nonlinear  $H_{\infty}$ -control problem (for the original plant P) in the internal state-space sense.

Under the assumption that  $a(x) = A(x) - B_1(x) D_{21}^{-1}$  $(x)C_2(x)$  is asymptotically stable and it is straightforward, by using the results of Section II, to compute a (j, J)-inner-outer factorization  $G = \Theta \circ R$  of G given by (59). The result is as follows.

Theorem 5: Let G be as in (59) and assume that  $a(x) := A(x) - B_1(x)D_{21}^{-1}(x)C_2(x)$  is an asymptotically stable vector field. Suppose that there exists a smooth solution P(x) of the Hamilton-Jacobi equation

$$P_{x}(x)[A(x) - B_{2}(x)E_{1}^{-1}(x)D_{12}^{T}(x)C_{1}(x)] \\ + \frac{1}{2}C_{1}(x)[I - D_{12}(x)E_{1}^{-1}(x)D_{12}^{T}(x)]C_{1}^{T}(x) \\ - \frac{1}{2}P_{x}(x)[B_{2}(x)E_{1}^{-1}(x)B_{2}(x) - \gamma^{-2}B_{1}(x) \\ \cdot B_{1}^{T}(x)]P_{x}^{T}(x) = 0, \qquad P(0) = 0$$
(60)

with  $P(x) \ge 0$  and with stabilizing side condition

$$A(x) - B_2(x)E_1^{-1}(x)D_{12}^T(x)C_1(x) - [B_2(x)E_1^{-1}(x)B_2(x) - \gamma^{-2}B_1(x)B_1^T(x)]P_x^T(x)$$
 is asymptotically stable  
(61)

where

$$E_1(x) = D_{12}^{T}(x)D_{12}(x).$$
 (62a)

Assume that  $E_1(x)$  in (62a) has a factorization

$$E_1(x) = \overline{d}_1^T(x)\overline{d}_1(x) \tag{62b}$$

for a smooth, square, and invertible function  $\overline{d}_1(x)$ . Then  $G = \Theta \circ R$  is a (j, J)-inner-outer factorization of G where

$$R: \begin{cases} \dot{x} = A(x) - B_1(x)D_{21}^{-1}(x)C_2(x) \\ + B_2(x)u + B_1(x)D_{21}^{-1}(x)y \\ u' = \overline{d}_1(x)E_1^{-1}(x)(D_{12}^T(x)C_1(x) + B_2^T(x)P_x^T(x)) \\ + \overline{d}_1(x)u \\ y' = -\gamma D_{21}^{-1}(x)C_2(x) - \gamma^{-1}B_1^T(x)P_x^T(x) \\ + \gamma D_{21}^{-1}(x)y \end{cases}$$
(63)

and

$$\Theta: \begin{cases} \dot{x} = [A(x) - B_2(x)E_1^{-1}(x)(D_{12}^T(x)C_1(x) \\ + B_2^T(x)P_x^T(x)) + \gamma^{-2}B_1(x)B_1^T(x)P_x^T(x)] \\ + B_2(x)\overline{d_1}^{-1}(x)u' + \gamma^{-1}B_1(x)y' \\ z = (I - D_{12}(x)E_1^{-1}(x)D_{12}^T(x))C_1(x) \\ - D_{12}(x)E_1^{-1}(x)B_2^T(x)P_x^T(x) \\ + D_{12}(x)\overline{d_1}^{-1}(x)u' \\ \tilde{w} = \gamma^{-1}B_1^T(x)P_x^T(x) + y'. \end{cases}$$
(64)

From the results of [9] (see [7], [18], [19], and [21] for the linear case), it is known that, under appropriate conditions, compensators  $K: y \to u$  solving the  $H_{\infty}$ -control problem for the plant P arise in the form

$$u = r_1(H(y'), y')$$
  

$$y = r_2(H(y'), y')$$
(65)

where  $(u', y') \rightarrow (u, y) = (r_1(u', y'), r_2(u', y'))$  defines the inverse  $(T_R)^{-1}$  of the input-output map  $T_R$  associated with the outer factor R in the (j, J)-inner-outer factorization  $G = \Theta \circ R$ , and where H is a free-parameter stable plant. The so-called "central compensator" corresponds to the choice H = 0 in (65). Thus the central compensator  $K_c: y \rightarrow u$  is specified by

$$T_R(K(y), y) = (0, y')$$

By imposing the constraint u' = 0 in (63) and solving for  $K_c: y \to u$ , we obtain the state-space equations for the central compensator  $K_c$ . The result is

$$K_{c}:\begin{cases} \hat{x} = A(\hat{x}) - B_{1}(\hat{x})D_{21}^{-1}(\hat{x})C_{2}(\hat{x}) \\ - B_{2}(\hat{x})E_{1}^{-1}(\hat{x})D_{12}^{T}(\hat{x})C_{1}(\hat{x}) \\ - B_{2}(\hat{x})E_{1}^{-1}(\hat{x})B_{2}^{T}(\hat{x})P_{x}^{T}(\hat{x}) \\ + B_{1}(\hat{x})D_{21}^{-1}(\hat{x})y \\ u = -E_{1}^{-1}(\hat{x})(D_{12}^{T}(\hat{x})C_{1}(\hat{x}) + B_{2}^{T}(\hat{x})P_{x}^{T}(\hat{x})). \end{cases}$$
(66)

We now discuss conditions for this compensator to be a solution of the  $H_{\infty}$ -control problem [for the original plant P given by (56)] in the internal state-space sense. If we close the loop [(56) combined with the compensator  $K_c: y \to u$  in (66)], we obtain as the closed-loop system

$$T_{c.\ell.}: \begin{cases} \dot{x} = A(x) - B_2(x)E_1^{-1}(\hat{x})(D_{12}^T(\hat{x})C_1(\hat{x}) \\ + B_2^T(\hat{x})P_x^T(\hat{x})) + B_1(x)w \\ \dot{\hat{x}} = A(\hat{x}) + B_1(\hat{x})D_{21}^{-1}(\hat{x})(C_2(x) - C_2(\hat{x})) \\ - B_2(\hat{x})E_1^{-1}(\hat{x})(D_{12}^T(\hat{x})C_1(\hat{x}) \\ + B_2^T(\hat{x})P_x^T(\hat{x})) + B_1(\hat{x})D_{21}(\hat{x})^{-1}D_{21}(x)w \\ z = C_1(x) - D_{12}(x)E_1^{-1}(\hat{x})(D_{12}^T(\hat{x})C_1(\hat{x}) \\ + B_2^T(\hat{x})P_x^T(\hat{x})). \end{cases}$$
(67)

The issue now is to show that the vector field

$$(x, \hat{x}) \rightarrow (A(x) - B_2(x)E_1^{-1}(x)(D_{12}^T(\hat{x})C_1(\hat{x}) + B_2^T(\hat{x})P_x^T(\hat{x})), A(\hat{x}) + B_1(\hat{x})D_{21}^{-1}(\hat{x}) \cdot (C_2(x) - C_2(\hat{x})) - B_2(\hat{x})E_1^{-1}(\hat{x}) \cdot (D_{12}^T(\hat{x})C_1(\hat{x}) + B_2^T(\hat{x})P_x^T(\hat{x})))$$
(68)

is asymptotically stable. Note that the diagonal (x,x) is invariant under the flow of this vector field. In fact, if the initial condition is taken on the diagonal and an arbitrary input signal w is fed in, then the state  $(x(t), \hat{x}(t))$  at time t remains on the diagonal  $x(t) = \hat{x}(t)$  for all t > 0, i.e., a minimal realization for the input-output map [where states are initialized at  $(x(0), \hat{x}(0)) = (0, 0)$ ] lies on the diagonal.

The connection of the solution P(x) of (60) and (61) with stability of the vector field (68) is given by the following result.

**Theorem 6:** If the solution P(x) of (60) and (61) is proper and positive definite, then the vector field

$$x \to A(x) - B_2(x)E_1^{-1}(x)(D_{12}^T(x)C_1(x) + B_2^T(x)P_x^T(x))$$
(69)

[the diagonal of the vector field (68)] is Lyapunov stable. If in addition the system

$$\dot{x} = A(x) - B_2(x)E_1^{-1}(x)(D_{12}^T(x)C_1(x) + B_2^T(x)P_x^T(x)) \begin{bmatrix} z \\ w \end{bmatrix} = \begin{bmatrix} (I - D_{12}(x)E_1^{-1}(x)D_{12}^T(x))C_1(x) \\ - D_{12}(x)E_1^{-1}(x)B_2^T(x)P_x^T(x) \\ \gamma^{-1}B_1^T(x)P_x^T(x) \end{bmatrix}$$
(70)

is detectable, then the vector field (69) is asymptotically stable.

#### BALL AND VAN DER SCHAFT: J-INNER-OUTER FACTORIZATION

*Proof:* To abbreviate the notation, write the realization for the system  $\Theta$  in (64) in the form

$$\Theta: \begin{cases} \dot{x} = A_{\Theta}(x) + B_{\Theta 1}(x)u' + B_{\Theta 2}(x)y' \\ z = C_{\Theta 1}(x) + D_{12}(x)\overline{d}_{1}^{-1}(x)u' \\ w = C_{\Theta 2}(x) + y'. \end{cases}$$
(71)

We know that  $\Theta$  is (j, J)-inner with storage function P, and hence [as can also be checked directly from (60) and (64)]

$$P_{x}(x)A_{\Theta}(x) + \frac{1}{2}C_{\Theta_{1}}^{T}(x)C_{\Theta_{1}}(x) - \frac{1}{2}C_{\Theta_{2}}^{T}(x)C_{\Theta_{2}}(x) = 0$$
(72)
$$P_{x}(x)[B_{\Theta_{1}}(x) \ B_{\Theta_{2}}(x)] + [C_{\Theta_{1}}^{T}(x) - C_{\Theta_{2}}^{T}(x)]$$

$$\cdot \begin{bmatrix} D_{12}(x)\overline{d}_{1}^{-1}(x) & 0\\ 0 & I \end{bmatrix} = 0.$$
(73)

From (72) and (73), we see that

$$P_x(x)(A_{\Theta}(x) - B_{\Theta 2}(x)C_{\Theta 2}(x)) = -\frac{1}{2}C_{\Theta 1}^T(x)C_{\Theta 1}(x) - \frac{1}{2}C_{\Theta 2}^T(x)C_{\Theta 2}(x).$$

The assertions of the theorem now follow from standard Lyapunov arguments, upon noting that  $A_{\Theta}(x) - B_{\Theta 2}(x)C_{\Theta 2}(x)$  coincides with the vector field (69) (see [16] and [23]).

The result of Theorem 6 is not satisfactory in that the ultimate goal is to prove the stability of the full vector field (68). For this purpose it is useful to consider the full system  $G \circ R^{-1}$  where G is given by (69), and R is given by (63). The resulting system is given by (31) after the appropriate substitutions

$$\begin{aligned} a(x) &= A(x) - B_1(x) D_{21}^{-1}(x) C_2(x) \\ b(x) &= \begin{bmatrix} B_2(x) & B_1(x) D_{21}^{-1}(x) \end{bmatrix} \\ c(x) &= \begin{bmatrix} C_1(x) \\ -\gamma D_{21}^{-1}(x) C_2(x) \end{bmatrix} \\ d(x) &= \begin{bmatrix} D_{12}(x) & 0 \\ 0 & \gamma D_{21}^{-1}(x) \end{bmatrix} \\ \bar{c}(x) &= \begin{bmatrix} \overline{d}_1(x) E_1^{-1}(x) (D_{12}^T(x) C_1(x) + B_2^T(x) P_x^T(x)) \\ -\gamma D_{21}^{-1}(x) C_2(x) - \gamma^{-1} B_1^T(x) P_x^T(x) \end{bmatrix} \\ \bar{d}(x) &= \begin{bmatrix} d_1(x) & 0 \\ 0 & D_{21}^{-1}(x) \end{bmatrix}. \end{aligned}$$
(74)

The input-state-output trajectories  $\{(w(t), x(t), \hat{x}(t), z(t))\}$ associated with the closed-loop system  $T_{c.\ell.}$  given by (67) coincides with the input-state-output trajectories  $\{((u'(t), y'(t)), (x(t), \hat{x}(t)), (z(t), w(t))\}$  of the system  $\Theta$  (64) subject to  $u'(t) \equiv 0$  and ignoring y'(t). If we write the system equations for  $G \circ R^{-1}$  in the form

$$G \circ R^{-1}: \begin{cases} \tilde{x} = A_{\hat{\Theta}}(\tilde{x}) + B_{\hat{\Theta}1}(\tilde{x})u' + B_{\hat{\Theta}2}(\tilde{x})y' \\ z = C_{\hat{\Theta}1}(\tilde{x}) + D_{12}(x)\overline{d}_1^{-1}(\hat{x})u' \\ w = C_{\hat{\Theta}2}(\tilde{x}) + D_{21}^{-1}(x)D_{21}(\hat{x})y' \end{cases}$$
(75)

[where  $\tilde{x} = (x, \hat{x})$ ], then the system equations for the closedloop system  $T_{c,\ell}$  in (67) can be written as

$$T_{c.\ell.}: \begin{cases} \dot{x} = A_{\hat{\Theta}}(\tilde{x}) - B_{\hat{\Theta}2}(\tilde{x})D_{21}^{-1}(\hat{x})D_{21}(x)C_{\hat{\Theta}2}(\tilde{x}) \\ + B_{\hat{\Theta}2}(\tilde{x})D_{21}^{-1}(\hat{x})D_{21}(x)w \\ z = C_{\hat{\Theta}1}(\tilde{x}). \end{cases}$$
(76)

If  $G \circ R^{-1}$  is (j, J)-lossless with storage function  $S(\tilde{x}) \ge 0$ , which is also proper and positive definite, then we can proceed exactly as in the proof of Theorem 5 to show that the state dynamics

$$\tilde{x} \to A_{\hat{\Theta}}(\tilde{x}) - B_{\hat{\Theta}2}(\tilde{x}) D_{21}^{-1}(\hat{x}) D_{21}(x) C_{\hat{\Theta}2}(\tilde{x}))$$

is Lyapunov stable and is, in fact, asymptotically stable under the appropriate detectability hypothesis. This leads to the following result. While the result is not definitive, it is comparable to those obtained elsewhere in the literature on this problem (see, e.g., [16], [17], [8], and [20]); in fact, the method of analysis which we have presented here has become the standard method for obtaining what stability results there are for this problem.

**Theorem 6:** Suppose that the system  $G \circ R^{-1}$  [equal to (31) with the substitutions (74)] is (j, J)-inner with storage function  $S(\tilde{x})$  ( $\tilde{x} = (x, \hat{x})$ ) which is positive definite and proper. Then the vector field (68) is Lyapunov stable. If in addition the system

$$\begin{split} \tilde{x} &= A_{\hat{\Theta}}(\tilde{x}) \\ \begin{bmatrix} z \\ w \end{bmatrix} = \begin{bmatrix} C_{\hat{\Theta}1}(\tilde{x}) \\ C_{\hat{\Theta}2}(\tilde{x}) \end{bmatrix} \end{split}$$

is detectable, then the vector field (68) is asymptotically stable and the compensator  $K_c$  as in (66) is a solution of the  $H_{\infty}$ -control problem for the plant P in (56) in the internal state-space sense.

*Remark:* Note, however, that if  $D_{21}(x)$  and/or  $D_{12}(x)$  is not constant, then the last of the conditions of (4)

$$\left(\text{with } d(\tilde{x}) = \begin{bmatrix} D_{12}(x)\overline{d}_1^{-1}(\hat{x}) & 0\\ 0 & D_{21}^{-1}(x)D_{21}(\hat{x}) \end{bmatrix} \right)$$

in general, holds only on the diagonal so  $G \circ R^{-1}$  cannot be (j, J)-inner. The result of Theorem 6 requires only the validity of the first two of (4) for  $G \circ R^{-1}$ . For purposes of verifying that  $K_c$  solves the  $H_{\infty}$ -control problem, it is sufficient that there exist a smooth, proper, positive definite storage function S with respect to which  $G \circ R^{-1}$  is (j, J)-dissipative. This in turn is somewhat stronger than just working directly with the condition that there be a smooth, positive definite, proper storage function  $S(\tilde{x})$  with respect to which  $T_{c.\ell.}$  is dissipative with respect to the supply rate  $w^T w - z^T z$ .

A related discussion of the connections between (j, J)inner-outer factorization and the  $H_{\infty}$ -control problem appears in [6].

In this paper we have shown how the disturbance feedforward case of the nonlinear  $H_{\infty}$ -control problem can be reduced to a *J*-inner-outer factorization problem which, in turn, can be reduced to solving a Hamilton-Jacobi equation with stabilizing side condition. Solving such a Hamilton-Jacobi equation in practice may be difficult; typically one cannot solve it explicitly and hence must resort to some approximate or numerical procedure. In [4] and [5], explicit calculations were carried out for the case where the plant is a memoryless system composed with a linear system. Here we offer another example where the presence of extra structure in the original plant enables one to find an explicit solution. IEEE TRANSACTIONS ON AUTOMATIC CONTROL, VOL. 41, NO. 3, MARCH 1996

*Example:* Consider the system

$$\Sigma: \begin{cases} \dot{x} = f(x) + g(x)u\\ y = h(x)) \end{cases}$$

where f and h are smooth with f(0) = 0 and h(0) = 0, u and y are in  $\mathbb{R}^m$ , and x is in the state manifold M. We assume that  $\Sigma$  is a lossless system in the sense that there exists a storage function  $H: M \to \mathbb{R}$  with H(0) = 0, H(x) > 0 for  $x \neq 0$ , such that we have

$$\frac{d}{dt}\{H(x(t))\} = u(t)^T y(t)$$

along trajectories (u(t), x(t), y(t)) of  $\Sigma$ , or equivalently

$$H_x(x)f(x) = 0, \quad H_x(x)g(x) = h^T(x)$$
 (77)

and that  $\Sigma$  is zero-state detectable, i.e.,  $\lim_{t\to\infty} x(t) = 0$ whenever y(t) = 0 for all  $t \ge 0$ ). The perturbation  $\Sigma_p$  of  $\Sigma$ based on the normalized stable kernel-representation of  $\Sigma$  (see [33]) is now given by

$$P: \begin{cases} \dot{x} = f(x) + g(x)u + g(x)w, & w \in \mathbb{R}^m \\ z = \begin{bmatrix} u \\ y \end{bmatrix} \\ y = g^T(x)H_x^T(x) + w. \end{cases}$$
(78)

The robust stabilization problem as considered in [33] is to construct a measurement feedback law  $K: y \to u$  to minimize the  $L_2$ -gain from w to  $z = \begin{bmatrix} u \\ y \end{bmatrix}$  for the resulting closed-loop system.

This problem can be solved by using the *J*-spectral factorization approach discussed in this section. With P as in (78), the associated system G in (59) becomes

$$G: \begin{cases} \dot{x} = [f(x) - g(x)g^T(x)H_x^T(x)] \\ + g(x)u + g(x)y \\ z = \begin{bmatrix} u \\ y \end{bmatrix} \\ \tilde{w} = -\gamma g^T(x)H_x^T(x) + \gamma y. \end{cases}$$
(79)

If we now make the substitutions

$$a(x) = f(x) - g(x)g^{T}(x)H_{x}^{T}(x) \quad b(x) = [g(x) \quad g(x)]$$

$$c(x) = \begin{bmatrix} 0 \\ 0 \\ -\gamma g^{T}(x)H_{x}^{T}(x) \end{bmatrix} \quad d(x) = \begin{bmatrix} 1 & 0 \\ 0 & 1 \\ 0 & \gamma \end{bmatrix}$$

$$J = \begin{bmatrix} I_{m} & 0 & 0 \\ 0 & I_{m} & 0 \\ 0 & 0 & -I_{m} \end{bmatrix} \quad E^{-1}(x) = \begin{bmatrix} 1 & 0 \\ 0 & (1 - \gamma^{2})^{-1} \end{bmatrix}$$
(80)

then (23) with side condition (24) reduces to

$$P_{x}(x)[f(x) - (1 - \gamma^{2})^{-1}g(x)g^{T}(x)H_{x}^{T}(x)] - \frac{1}{2}[(1 - \gamma^{2})^{-1} + 1]P_{x}(x)g(x)g^{T}(x)P_{x}^{T}(x) - \frac{1}{2}\gamma^{2}(1 - \gamma^{2})^{-1}H_{x}(x)g(x)g^{T}(x)H_{x}^{T}(x) = 0$$
(81)

with stability side condition

Ì

$$f - (1 - \gamma^2)^{-1} g g^T H_x^T - (2 - \gamma^2) (1 - \gamma^2)^{-1} g g^T P_x^T$$
  
is asymptotically stable. (81')

This is exactly the Hamilton–Jacobi equation (29) considered in [33]. Equation (81) has solution

$$P(x) = \gamma^{2} (\gamma^{2} - 2)^{-1} H(x)$$

for  $\gamma > \sqrt{2}$ . The assumption that  $\Sigma$  is zero-state detectable implies that P also meets the stability side condition (81'). Moreover it can be shown that the infimum of the set of  $\gamma$ 's for which a solution exists is  $\gamma^* = \sqrt{2}$  (see [33]). From (27)–(29) with the substitutions of (80) (or from (63) with the appropriate substitutions) we obtain, for  $\gamma > \sqrt{2}$ , the equation shown at the bottom of the page with  $\overline{d}(x) = \begin{bmatrix} 1 & 0 \\ 0 & \sqrt{\gamma^2 - 1} \end{bmatrix}$  if we take  $j = \begin{bmatrix} I_m & 0 \\ 0 & -I_m \end{bmatrix}$ . The central compensator is obtained by setting u' = 0; this yields

$$K_c: \begin{cases} \dot{x} = f(\hat{x}) - g(\hat{x})g^T(\hat{x})H_x^T(\hat{x}) \\ &- \gamma^2(\gamma^2 - 2)^{-1}g(\hat{x})g^T(\hat{x})H_x^T(\hat{x}) + g(\hat{x})y \\ &= [f(\hat{x}) - \gamma^2(\gamma^2 - 2)^{-1}g(\hat{x})g^T(\hat{x})H_x^T(\hat{x})] \\ &+ g(\hat{x})[y - g^T(\hat{x})H_x^T(\hat{x})] \\ &u = -\gamma^2(\gamma^2 - 2)^{-1}g^T(\hat{x})H_x^T(\hat{x}). \end{cases}$$

This is precisely the controller obtained via the certainty equivalence principle in [33]!

Based on the linear theory, the same controller for the general "disturbance feedforward problem" has been proposed in [20], where it has been also shown that this controller solves the suboptimal  $H_{\infty}$ -control problem at least locally.

### V. THE UNSTABLE CASE

In this section we discuss the difficulties of our approach to the (j, J)-inner-outer factorization problem in the case where the vector field a in (1) for the system  $\Sigma$  is not stable.

Let us assume that we are given a system  $\Sigma$  with statespace realization as in (1) for which the vector field a is not necessarily stable. Then it is still the case that the system  $[D\Sigma]^T \circ J\Sigma$  can be formed and has the Hamiltonian realization as in (12) with Hamiltonian H(x, p, u) given by (13) and that the inverse system  $([D\Sigma]^T \circ J\Sigma)^{-1}$  has Hamiltonian realization (20) with Hamiltonian  $H^{\times}$  given by (22). To write the system in the form  $[DR]^T \circ jR$  with R outer, the first step in our procedure is to find a new choice of canonical coordinates  $(\overline{x}, \overline{p})$  so that the antistable invariant manifold for the *H*-Hamiltonian flow (with u = 0) is given by  $\{(\overline{x}, \overline{p}): \overline{x} = 0\}$ , and that the stable invariant manifold for the  $H^{\times}$ -Hamiltonian flow (with  $y_a = 0$ ) is given by  $\{(\overline{x},\overline{p}): \overline{p} = 0\}$ . Indeed it is always possible to find local coordinates  $\overline{x}, \overline{p}$  in this way (see [1]). We can still proceed as before to find new canonical coordinates  $(\overline{x}, \overline{p})$  so that

$$R: \begin{cases} \dot{x} = [f(x) - g(x)g^{T}(x)H_{x}^{T}(x)] + g(x)u + g(x)y \\ \begin{bmatrix} u' \\ y' \end{bmatrix} = \overline{d}(x) \left( \begin{bmatrix} \gamma^{2}(\gamma^{2} - 2)^{-1}g^{T}(x)H_{x}^{T}(x) \\ \gamma^{2}(\gamma^{2} - 2)^{-1}(\gamma^{2} - 1)^{-1}g^{T}(x)H_{x}^{T}(x) \end{bmatrix} + \begin{bmatrix} u \\ y \end{bmatrix} \right)$$

390

the stable invariant manifold for the  $H^{\times}$ -Hamiltonian flow (with  $y_a = 0$ ) is equal to the manifold  $\{(\overline{x}, \overline{p}): \overline{p} = 0\}$ by solving(23) with the stability side condition (24). The remaining difficulty is that the antistable invariant manifold for the *H*-Hamiltonian flow (with u = 0) is not necessarily equal to  $\{(\overline{x}, \overline{p}): \overline{x} = 0\}$ , so that the system *R* defined by (29), in general, is not stable.

To conserve notation, let us assume that we already have this situation at the start. Thus we are given a system  $\Sigma$  as in (1), and Assumption (A2) is replaced by

$$p = m$$
 and  $d(x)$  is invertible for all  $x$  (A2')

and (A1) is replaced by

$$x_0 = 0$$
 is a globally asymptotically stable equilibrium  
point of  $\dot{x} = a(x) - b(x)d^{-1}(x)c(x)$ . (A1')

With these assumptions in place, it is clear that the stable invariant manifold for the  $H^{\times}$ -Hamiltonian flow (with  $y_a = 0$ ) is  $\{(x, p): p = 0\}$ . We seek a new choice of canonical coordinates  $(\overline{x}, \overline{p})$  so that the stable invariant manifold of the  $H^{\times}$ -Hamiltonian flow is still of the form  $\{(\overline{x}, \overline{p}): \overline{p} = 0\}$  but also the antistable invariant manifold for the *H*-Hamiltonian flow is equal to  $\{(\overline{x}, \overline{p}): \overline{x} = 0\}$ .

We assume that the antistable invariant manifold has the form  $\mathcal{M} = \{(P_p(p), p): (x, p) \in T^*M \text{ for some } x\}$  for a smooth function  $p \to P(p)$ . This leads to a Hamilton-Jacobi equation of the form

$$H(P_p^T(p), p, 0) = 0)$$

with stability side condition

- ---

$$-\frac{\partial H}{\partial x}(P_p^T(p), p, 0)$$
 is an antistable vector field.

Explicitly, the Hamilton–Jacobi equation is

$$p^{T}a(P_{p}^{T}(p)) + \frac{1}{2}[c(P_{p}^{T}(p))]^{T}jc(P_{p}^{T}(p)) = 0$$
 (82)

with stability side condition

$$\left[\frac{\partial a}{\partial x}(P_p^T(p))\right]^T p + \left[\frac{\partial c}{\partial x}(P_p(p))\right]^T jc(P_p^T(p)) \text{ is stable.}$$
(83)

Assume that we can solve (82) with the side condition (83). Then the new canonical set of coordinates  $(\bar{x}, p)$  given by  $(x, p) = (\bar{x} + P_p^T(p), p)$  has all the required properties and Step 1 of our procedure is complete. The Hamiltonian in the new coordinates  $(\bar{x}, p)$  becomes

$$\overline{H}(\overline{x}, p, u) = H(\overline{x} + P_p^T(p), p, u)$$

$$= p^T[a(\overline{x} + P_p^T(p)) + b(\overline{x} + P_p^T(p))u]$$

$$+ \frac{1}{2}[c(\overline{x} + P_p^T(p)) + d(\overline{x} + P_p^T(p))u]^T$$

$$\cdot j[c(\overline{x} + P_p^T(p)) + d(\overline{x} + P_p^T(p))u]. \quad (84)$$

The goal now is to identify functions  $\overline{a}(\overline{x}), \overline{b}(\overline{x}), \overline{c}(\overline{x}), \overline{d}(\overline{x})$  so

that  $\overline{H}$  has the form of (16), i.e., so that

$$\overline{H}(\overline{x}, p, u) = p^{T}[\overline{a}(\overline{x}) + \overline{b}(\overline{x})u] + \frac{1}{2}[\overline{c}(\overline{x}) + d(\overline{x})u]^{T}j[\overline{c}(\overline{x}) + \overline{d}(\overline{x})u].$$
(85)

Note that p appears linearly in (85) while the dependence on p of  $\overline{H}(\overline{x}, p, u)$  in (84) is a complicated nonlinear dependence. This yields a fundamental obstruction to finding  $\overline{a}, \overline{b}, \overline{c}, \overline{d}$  so that (85) matches (84).

The analysis here hinges on the demand that  $[D\Sigma]^T \circ J\Sigma$ match with  $[DR]^T \circ jR$  at the state-space level. As has been suggested by Helton, conceivably there is an R with a state-space dimension larger than that of  $\Sigma$  for which  $[D\Sigma]^T \circ J\Sigma = [DR]^T \circ jR$  at least at the input-output level. This issue of existence of (j, J)-inner-outer factorization in some sense for a unstable system  $\Sigma$  remains a current area of research (see, e.g., [36]).

#### REFERENCES

- V. I. Arnold, Mathematical Methods of Classical Mechanics, 2nd ed. Berlin: Springer-Verlag, 1989.
   J. A. Ball and J. W. Helton, "Shift invariant manifolds and nonlinear
- [2] J. A. Ball and J. W. Helton, "Shift invariant manifolds and nonlinear analytic function theory," *Integral Equations Operator Theory*, vol. 11, pp. 615–725, 1988.
- [3] "Factorization and general properties of nonlinear Toeplitz operators," in *The Gohberg Anniversary Collection*, vol. II, H. Dym, S. Goldberg, M. A. Kaashoek, and P. Lancaster, Eds. Basel: Birkhauser-Verlag, 1989, pp. 25–41.
- [4] \_\_\_\_\_, "Inner-outer factorization of nonlinear operators," J. Functional Anal., vol. 104, pp. 363–413, 1992.
   [5] \_\_\_\_\_, "Nonlinear H<sup>∞</sup> control theory for stable plants," Math. Contr.,
- [5] \_\_\_\_\_, "Nonlinear H<sup>∞</sup> control theory for stable plants," Math. Contr., Signals Syst., vol. 5, pp. 233-261, 1992.
  [6] \_\_\_\_\_, "Nonlinear H<sup>∞</sup>-control and J-inner-outer factorization," sub-
- [6] \_\_\_\_\_, "Nonlinear  $H^{\infty}$ -control and J-inner-outer factorization," submitted for publication.
- [7] J. A. Ball, J. W. Helton, and M. Verma, "A factorization principle for stabilization of linear control systems," *Int. J. Robust Nonlinear Contr.*, vol. 1, pp. 229–294, 1991.
- [8] J. A. Ball, J. W. Helton and M. L. Walker, "H<sup>∞</sup>-control for nonlinear systems with output feedback," *IEEE Trans. Automat. Contr.*, vol. 38, pp. 546–559, 1993.
  [9] J. A. Ball and M. Verma, "Factorization and feedback stabilization for
- [9] J. A. Ball and M. Verma, "Factorization and feedback stabilization for nonlinear systems," Syst. Contr. Lett., vol. 23, pp. 187–196, 1991.
  [10] H. Bart, I. Gohberg, and M. A. Kaashoek, Minimal Factorization of
- [10] H. Bart, I. Gohberg, and M. A. Kaashoek, *Minimal Factorization of Matrix and Operator Functions*. Basel: Birkhauser-Verlag, 1979.
  [11] T. Basar and P. Bernhard, H<sup>∞</sup>-Optimal Control and Related Minimax
- [11] T. Basar and P. Bernhard, H<sup>∞</sup>-Optimal Control and Related Minimax Design Problems: A Dynamic Game Approach. Basel: Birkhauser-Verlag, 1991.
- [12] P. E. Crouch and A. J. van der Schaft, Variational and Hamiltonian Control Systems. Basel: Springer-Verlag, 1987.
  [13] F. J. Doyle, F. Allgoewer, and M. Morari, "A normal form approach
- [13] F. J. Doyle, F. Allgoewer, and M. Morari, "A normal form approach to approximate input-output linearization for maximum phase nonlinear systems," *IEEE Trans. Automat. Contr.*, to appear.
  [14] M. Green, "H<sub>∞</sub>-controller synthesis by J-lossless coprime factoriza-
- [14] M. Green, "H<sub>∞</sub>-controller synthesis by J-lossless coprime factorizations," SIAM J. Contr. Opt., vol. 30, pp. 522–547, 1992.
  [15] M. Green, K. Glover, D. J. N. Limebeer, and J. C. Doyle, "A J-spectral
- [15] M. Green, K. Glover, D. J. N. Limebeer, and J. C. Doyle, "A J-spectral factorization approach to H<sub>∞</sub>-control," SIAM J. Contr. Opt., vol. 28, pp. 1350–1371, 1990.
- [16] A. Isidori and A. Astolfi, "Nonlinear  $H_{\infty}$  control via measurement feedback," J. Math. Syst., Estimation Contr., vol. 2, pp. 31–35.
- [17] \_\_\_\_\_, "Disturbance attenuation and H<sub>∞</sub>-control via measurement feedback in nonlinear systems," *IEEE Trans. Automat. Contr.*, vol. 37, pp. 1283–1293, 1992.
  [18] H. Kimura, "Chain-scattering representation, J-lossless factorization
- [18] H. Kimura, "Chain-scattering representation, J-lossless factorization and H<sup>∞</sup> control," J. Math. Syst., Estimation, Contr., vol. 4, pp. 401–450, 1994.
- [19] R. Kondo and S. Hara, "On cancellation in H<sup>∞</sup> optimal controllers," Syst. Contr. Lett., vol. 13, pp. 205–210, 1989.
  [20] W.-M. Lu and J. C. Doyle, "H<sub>∞</sub> control of nonlinear systems: A class
- [20] W.-M. Lu and J. C. Doyle, " $H_{\infty}$  control of nonlinear systems: A class of controller," California Institute of Technology, Tech. Memo 93-008, 1993.

#### IEEE TRANSACTIONS ON AUTOMATIC CONTROL, VOL. 41, NO. 3, MARCH 1996

- [21] M.-C. Tsai and I. Postlethwaite, "On J-lossless coprime factorizations and  $H^{\infty}$  control," Int. J. Robust Nonlinear Contr., vol. 1, pp. 47-68, 1991
- [22] A. J. van der Schaft, "On a state space approach to nonlinear  $H_{\infty}$ control," Syst. Contr. Lett., vol. 16, pp. 1–8, 1991. \_\_\_\_\_, " $L_2$ -gain analysis of nonlinear systems and nonlinear state
- [23] feedback  $H_{\infty}$  control," IEEE Trans. Automat. Contr., vol. 37, pp. 770-784, 1992.
- , "Nonlinear state space  $H_\infty$  control theory," in Essays on Con-[24] trol: Perspectives in the Theory and its Applications, H. L. Trentelman
- and J. C. Willems, Eds. Basel: Birkhauser-Verlag, 1993, pp. 153-190. [25] A. J. van der Schaft and J. A. Ball, "Inner-outer factorization of nonlinear state space systems," in Systems and Networks: Mathematical Theory and Applicatons, vol. II, U. Helmke, R. Mennicken, and J. Sauer, Eds. Berlin: Akademie Verlag, 1994, pp. 53-56.
- [26] "Nonlinear inner-outer factorization," in Proc. 33rd Conf. Decision Contr., Lake Buena Vista, FL, 1994, pp. 2549-2552.
- [27] in preparation. [28] M. Vidyasagar, Nonlinear Systems Analysis. San Diego: Prentice-Hall, 1993.
- [29] J. C. Willems, "Dissipative dynamical systems, Part I: General theory,"
- Arch. Rat. Mech. Anal., vol. 45, pp. 321–351, 1972.
  [30] R. A. Wright and C. Kravaris, "Nonminimum-phase compensation for nonlinear processes," AIChE J., vol. 38, pp. 26-40, 1992. [31] M. Berger, Nonlinearity and Functional Analysis: Lectures on Nonlinear
- Problems in Functional Analysis. New York: Academic, 1977.
- [32] D. J. Hill and P. J. Moylan, "Connections between finite gain and asymptotic stability," IEEE Trans. Automat. Contr., vol. 25, pp. 931-936, 1980.
- [33] A. J. van der Schaft, "Robust stabilization of nonlinear systems via stable kernel representations with  $L_2$ -gain bounded uncertainty," Syst. Contr. Lett., vol. 24, pp. 75–81, 1995. [34] M. Vidyasagar and A. Vannelli, "New relationships between input-
- output and Lyapunov stability," IEEE Trans. Automat. Contr., vol. 27, [35] J. L. Wyatt, L. O. Chua, J. W. Gannett, I. C. Goknar, and D. N. Green,
- "Energy concepts in the state-space theory of nonlinear n-ports: Part II-Losslessness," IEEE Trans. Circuits Syst., vol. 29, pp. 417-430, 1982
- [36] J. W. Helton and M. R. James, "An information approach to nonlinear J-inner-outer factorization," in Proc. 33rd Conf. Decision Contr., Lake Buena Vista, FL, 1994, pp. 2565-2571.



Joseph A. Ball (M'90) received the Ph.D. degree from the University of Virginia, Charlottesville, in 1973.

In 1973 he joined the faculty of the Department of Mathematics at Virginia Polytechnic Institute, Blacksburg. He has had visiting positions at the University of California, San Diego, the Weizmann Institute of Science in Israel, the Vrije Universiteit in Amsterdam, and the Mathematical Sciences Research Institute in Berkeley. His research interests include operator theory, the theory of matrix and

operator-valued functions, and their applications to system theory and control. He is coauthor of the book Interpolation of Rational Matrix Functions (Basel: Birkhauser, 1990). He is on the editorial board of Integral Equations and Operator Theory, the Journal of Mathematical Systems, Estimation and Control, and the Journal of Mathematical Analysis and its Applications.



Arjan J. van der Schaft (M'91) was born in Vlaardingen, The Netherlands, in February 1955. He received the undergraduate and Doctorate degrees in mathematics from the University of Groningen, The Netherlands, in 1979 and 1983, respectively.

In 1982 he joined the Department of Applied Mathematics, University of Twente, Enschede, The Netherlands, where he is presently an Associate Professor. His research interests include the analysis and control of nonlinear and mechanical systems and the mathematical modeling of physical systems.

He is the author of System Theoretic Descriptions of Physical Systems (CWI: 1984) and the coauthor of Variational and Hamiltonian Control Systems (Berlin: Springer-Verlag, 1987) and Nonlinear Dynamical Control System (Berlin: Springer-Verlag, 1990).